Author’s response to reviews

Title: The determinants and consequences of adult nurse staff turnover: a systematic review of systematic reviews

Authors:

Mary Halter (maryhalter@sgul.kingston.ac.uk; maryhalter@hotmail.co.uk)
Olga Boiko (olga.boiko1@gmail.com)
Ferruccio Pelone (f.pelone@sgul.kingston.ac.uk)
Carole Beighton (c.beighton@sgul.kingston.ac.uk)
Ruth Harris (ruth.harris@kcl.ac.uk)
Julia Gale (j.gale@sgul.kingston.ac.uk)
Stephen Gourlay (gourlaysn@kingston.ac.uk)
Vari Drennan (v.m.drennan@sgul.kingston.ac.uk)

Version: 2 Date: 27 Sep 2017

Author’s response to reviews:

Dear Dr Logan,

Thank you for your email and acceptance of this paper for publication.

As per your instruction we have not uploaded a cover letter, but have changed the title of the Ethics approval section to Ethics approval and consent to participate and included a Consent to publish section in our Declarations. We have also removed all tracked changes, comments, highlights and text in different colours in the manuscript and supplementary material.

Best wishes

Mary Halter